**Author’s response to reviews**

**Title:** What Really Happens in the Home: A Comparison of Parent-Reported and Observed Tooth Brushing Behaviors for Young Children

**Authors:**

Molly Martin (mollyma@uic.edu)

Genesis Rosales (grosal2@uic.edu)

Anna Sandoval (asando1@uic.edu)

Helen Lee (leehelen@uic.edu)

Oksana Pugach (opugach@uic.edu)

David Avenetti (avenetti@uic.edu)

Gizelle Alvarez (galvar22@uic.edu)

Anabelen Diaz (adiaz45@uic.edu)

**Version:** 3 **Date:** 06 Feb 2019

**Author’s response to reviews:**

January 21, 2019

Regarding: Submission to BMC Oral Health - OHEA-D-18-00428

Dear Dr. Bhoopathi,

Thank you for the tentative acceptance of our manuscript. We completed the changes you requested, as explained below.

Sincerely,

The Authors
1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Response: The emails have been added.

2. Please rename Objective to Background.

Response: In the abstract, we changed Objective to Background.

3. Please rename Introduction to Background.

Response: In the main text, we changed Introduction to Background.